Author's response to reviews

Title: Seasonal patterns of suicides over the period of socio-economic transition in Lithuania

Authors:
- Ramune Kalediene (kaleda@kaunas.omnitel.net)
- Skirmante Starkuviene (smante@centras.lt)
- Jadvyga Petrauskiene (socmed@kmu.lt)

Version: 5 Date: 7 February 2006

Author's response to reviews: see over
7 February 2006

MS: 9962259748157860
Seasonal patterns of suicides over the period of socio-economic transition in Lithuania
Ramune Kalediene, Skirmante Starkuviene and Jadvyga Petrauskiene
BMC Public Health

Following formatting changes for the manuscript were done:
1. Ensured that the title and abstract were identical in the manuscript and in the submission system.
2. Ensured that the authors list in the manuscript written identically and in the same order they are in the submission system.
3. The sentence case (all unnecessary capitals removed) used in the title, both in the submission system and in the manuscript.
4. Abstract structured into required sections and updated on the submission page.
5. Any highlighting and tracking removed from the manuscript text.
6. Ethical considerations moved to the Methods section.
7. Figure numbers and titles removed from the figure files.
8. Figures cropped as closely as possible, according to the rules presented in the cropping guide.

We hope that with your support, we achieved the required level of the paper to be published in your journal and look forward to further information from you.

Sincerely,
Authors